Author's response to reviews

Title: Impact on CAesarean section Rates following Injections of Sterile water (ICARIS): A multicentre randomised trial.

Authors:

Nigel Lee (Nigel.lee@mater.org.au)
Lena B Mårtensson (Lena.Martensson@his.se)
Caroline Homer (Caroline.Homer@uts.edu.au)
Joan Webster (Joan.Webster@health.qld.gov.au)
Kristen Gibbons (kgibbons@mmri.mater.org.au)
Helen Stapleton (Helen.Stapleton@mater.org.au)
Natalie Dos Santos (Natalie.dossantos@mater.org.au)
Michael Beckmann (Michael.Beckmann@mater.org.au)
Yu Gao (Yu.Gao@ucrh.edu.au)
Sue Kildea (suekildea@acu.edu.au)

Version: 3 Date: 3 April 2013

Author's response to reviews: see over
Emily Crow
Executive Editor
BioMedCentral: Pregnancy and Childbirth
Floor 6, 236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Ms Crow,
The amended manuscript for the ICARIS trial protocol has been uploaded to the BMC site as requested. The amended protocol contains the changes that you requested regarding limiting reference to the six sites that we currently have ethics approval for. Many thanks for your patience in this matter; I trust that you will find the changes suitable.

Sincerely,

Nigel Lee
ICARIS Trial – Project Manager
Australian Catholic University
Mater Medical Research Institute
Level 2 Aubigny Place
South Brisbane Qld 4101
AUSTRALIA

Permission to reprint Figure 2 was obtained from author, Sally Tracy.